Author's response to reviews

Title: Evaluation of antibacterial and cytotoxic activity of Artemisia nilagirica and Murraya koenigii leaf extracts against mycobacteria and macrophages

Authors:

Sumanta Kumar Naik (sknbiotech@gmail.com)
Soumitra Mohanty (soumitra.dna@gmail.com)
Avinash Padhi (avinashpadhi134@gmail.com)
Rashmirekha Pati (rashmi_pati@kiitbiotech.ac.in)
Avinash Sonawane (asonawane@kiitbiotech.ac.in)

Version: 4
Date: 16 February 2014

Author's response to reviews: see over
To
Prof. Tom Rowles
Executive Editor
BMC Complementary and Alternative Medicine

Dear Prof. Tom Rowles,

As suggested by you, herewith we are submitting a revised manuscript (Ref. MS: 1708304829111889) entitled “Evaluation of antibacterial and cytotoxic activity of Artemisia nilagirica and Murraya koenigii leaf extracts against mycobacteria and macrophages”, to consider for publication in BMC Complementary and Alternative Medicine journal. In this revised manuscript, we have addressed the Editorial comments. Reviewers had no comments. The responses to the Editorial comments are given below.

The manuscript has been approved by all authors and is not being submitted concurrently elsewhere.

Yours sincerely,

Avinash Sonawane
Responses to Editorial comments:

Editorial comment:

Query 1: Please include an Authors' Contributions section in your manuscript.

Answer: In the revised manuscript, we have included the Authors Contribution Section.

Q2: Please also ensure that your revised manuscript conforms to the journal style

Answer: The revised manuscript conforms the journal style.